Author's response to reviews

Title: Association of health behaviour with heart rate variability: a population-based study

Authors:

Alexander Kluttig (alexander.kluttig@medizin.uni-halle.de)
Barbara Schumann (barbara.schumann@epiph.umu.se)
Cees A Swenne (C.A.Swenne@lumc.nl)
Jan A Kors (j.kors@erasmusmc.nl)
Oliver Kuß (oliver.kuss@medizin.uni-halle.de)
Hendrik Schmidt (hendrik.schmidt@klinikum-magdeburg.de)
Karl Werdan (karl.werdan@medizin.uni-halle.de)
Johannes Haerting (johannes.haerting@medizin.uni-halle.de)
Karin H Greiser (h.greiser@Dkfz-Heidelberg.de)

Version: 3 Date: 25 November 2010

Author's response to reviews: see over
Dear Sir or Madam,

We would like to submit the revised version of the manuscript “Association of Health Behaviour with Heart Rate Variability: A Population-Based Study”. We respond to your formatting change requests indicated in your email of November 24 as follows:

Author list: we checked that the author lists are identical in manuscript and submission system.

Title: requested changes have been made in the revised manuscript and submission system (capitalized only the first word).

Authors' contributions: we added a sentences indicating that all authors read and approved the final manuscript. We formatted these in to a paragraph.

Highlighting: we removed all highlighting from the manuscript.

Abbreviations: we formatted these in to a paragraph.

Figures: we cropped all figures as closely as possible to minimise white space around the images. All figures are in the correct orientation.

Typography: we checked our manuscript for any typographical errors.

I hope that we completed all requested changes to your satisfaction.

Thank you very much for your efforts.

Sincerely,

Alexander Kluttig (corresponding author)